Author's response to reviews

Title: Computer Simulation in Conjunction with Medical Thermography as an Adjunct Tool for Early Detection of Breast Cancer

Authors:

   Prof Eddie Y.K. Ng (mykng@ntu.edu.sg)
   Natteri M Sudharsan (sudharsann@asme.org)

Version: 4 Date: 15 Mar 2004

Dear Sir,
Thanks for your comments on the script.
We have:
1. removed the authors' job titles as suggested.
2. ensured the format on list of refs. with the suggested style.
3. Cropped the Figs. 1 to 4 in JPEG format. These images have been cropped to the best possible. In figures 2,3 & 4 we are comparing numerical with experimental, the experimental plate is smaller than the numerical one and hence some white space above experimental image is unavoidable. But I hope that this is acceptable, and this is the best we can crop.
Thanks & regards,
EYK Ng